Author’s response to reviews

Title: EFFECTS OF CYCLIC PARENTERAL NUTRITION ON PARENTERAL-ASSOCIATED LIVER DYSFUNCTION PARAMETERS

Authors:
Jose Javier Arenas Villafranca (jjavier.arenas@gmail.com)
Miriam Nieto Guindo (miriamnietog@hotmail.com)
Elena Álvaro Sanz (elena.as84@hotmail.com)
Manuela Moreno Santamaria (manuelamorenosantamaria@gmail.com)
Marga Garrido Siles (marga.garrido@hotmail.com)
Jimena Abilés (jimesolea@yahoo.es)

Version: 1 Date: 11 Jan 2017

Author’s response to reviews:
Following text has been modified in manuscript

"Ethics approval and consent to participate:

This study was approved by the Medical Ethics Committee of Costa del Sol Hospital (CIF:7955060-D), this Committee also gave approval to the exemption of informed consent, since this study was retrospective study whose information management was always done respecting rules in force. All of the protocols and procedures were performed according to the Declaration of Helsinki".